Author’s response to reviews

Title: Renal capsule metastasis from renal pelvic cancer: a case report

Authors:

Yasuyuki Kobayashi (ya_su_koba@yahoo.co.jp)
Hiroki Arai (arai_h@kich.itami.hyogo.jp)
Masahito Honda (honda_m@kich.itami.hyogo.jp)
Takashi Matsumoto (matsumoto_t@kich.itami.hyogo.jp)
Kyotaro Yoshida (yoshida_k@kich.itami.hyogo.jp)

Version: 1 Date: 31 Oct 2017

Author’s response to reviews:

Dear editor,

Thank you for sending me the mail.

I received your mail about the paper 'Renal capsule metastasis from renal pelvic cancer: a case report'.

Regarding your inquiry, my response is as follows.

1) If you wish to make any changes to authorship before you resubmit your revisions, please reply to this email and ask for a 'Request for change in authorship' form which should be completed by all authors (including those to be removed) and returned to this email address.

→ I do not want to make any changes to the authors and order of authors.

2) We had noticed only five of the eight authors are mentioned in this section.

Please revise this section to include all authors, so that all authors listed in the manuscript can qualify for authorship.

The authors are only five (Yasuyuki Kobayashi; Hiroki Arai; Masahito Honda; Takashi Matsumoto; Kyotaro Yoshida).
3) We think we can find reference 10 in the PUBMED.


HER2 protein overexpression and gene amplification in upper urinary tract urothelial carcinoma-an analysis of 171 patients.


Please let me know if there is a problem with the notation.

Thank you very much for your support of my paper.

Best regards,

Yasuyuki Kobayashi.

Department of Urology, Kinki Central Hospital
3-1 Kurumaduka, Itami city, Hyogo 664-8533, Japan
Phone: +81-72-781-3712
Fax: +81-72-779-1567